Author's response to reviews

Title: Obesity: An overview on its current perspectives and treatment options

Authors:

Dr Srinivas Nammi (nammi@rediffmail.com)
Saisudha Koka (k_saisudha@yahoo.com)
Krishna M. Chinnala (chinnala_km@yahoo.com)
Dr Krishna M. Boini (krishnamurthyboini@yahoo.com)

Version: 2 Date: 30 Mar 2004

To
The Editor,
Nutrition Journal

MS: 1210417957308661

Dear Sir,

Thank you very much for accepting our article for publication in Nutrition Journal. Please find attached the text and figure files after re-formatting as per your suggestions.

I look forward to hearing from you.

With regards,

Sincerely,

Srinivas Nammi